Author's response to reviews

Title: Integrated Personal Health Records: Transformative Tools for Consumer-Centric Care

Authors:

Don E. Detmer (detmer@amia.org)
Meryl Bloomrosen (meryl@amia.org)
Brian Raymond (Brian.Raymond@kp.org)
Paul Tang (pctang@pacbell.net)

Version: 4 Date: 27 August 2008

Author's response to reviews:

To the Editor,

We have made your requested changes to help ensure that the manuscript conforms to all of the reviewer points. We revised the abstract and the manuscript sections. We have addressed author competing interests.

Sincerely,
Meryl Bloomrosen